Author's response to reviews

Title: Evaluation and treatment of low and anxious mood in Chinese speaking international students studying in Scotland: Study protocol of a pilot randomised controlled trial.

Authors:

Zheng Mengyi (m.zheng.1@research.gla.ac.uk)
Carrie-Anne McClay (c.mcclay.1@research.gla.ac.uk)
Sarah Wilson (sarah.wilson@glasgow.ac.uk)
Chris Williams (chris.williams@glasgow.ac.uk)

Version: 4 Date: 20 May 2015

Author's response to reviews: see over
Dear Sir/Madam,

Thank you for the opportunity to respond to the editor’s and reviewers’ comments regarding the manuscript: MS: 3885608611484322, “Evaluation and treatment of low and anxious mood in Chinese speaking international students studying in Scotland: Study protocol of a pilot randomised controlled trial“.

We found them very helpful and have amended the manuscript accordingly. The manuscript has been read and amended by a native English speaker and justification for the 6 month follow-up has been added on P.9.

I look forward to hearing your final decision.

Yours sincerely,

Professor Chris Williams
Tel: +44 (0)141 211 9123
Email: chris.williams@glasgow.ac.uk